Author’s response to reviews

Title: A Single-Center Retrospective Study of Factors Related to the Effects of Intravenous Glucocorticoid Therapy in Moderate-To-Severe and Active Thyroid-associated Ophthalmopathy.

Authors:

Yang Wang (ocean_wy@foxmail.com)
Shuo Zhang (akaseki@126.com)
Xingtong Liu (liuxingtong124@163.com)
Yidan Zhang (zhangyd513@126.com)
Hao Gu (gujinping1978@163.com)
Sisi Zhong (sisizhong916@163.com)
Yazhuo Huang (colourful4210@foxmail.com)
Sijie Fang (fangsijie89@hotmail.com)
Jing Sun (sophiasj@126.com)
Huifang Zhou (fangzzfang@163.com)
Xianqun Fan (fanxq@sjtu.edu.cn)

Version: 3 Date: 07 Feb 2018

Author’s response to reviews:

Feb 8, 2018

Dear Editor,

We are grateful for your encouraging remarks regarding our manuscript entitled: “A Single-Center Retrospective Study of Factors Related to the Effects of Intravenous Glucocorticoid Therapy in Moderate-To-Severe and Active Thyroid-associated Ophthalmopathy.”

According to the remarks, we modified our manuscript including:
1. We have clearly stated that the written informed consents were obtained from all participants. (Page 18 Line 6)

2. We have included the email addresses for all authors on the title page. (Page 1 Line 15)

3. SZ, YZ, XL, HG, SZ, YH, and SF not only performed the ophthalmic assessments but also made substantial contributions to acquisition of data. Data collection was heavy work in our study. We believed they were qualified for authorship. (Page 19 Line 13)

4. We have uploaded our manuscript, table and figures as clean versions without any track changes.

We hope that the revised version of the manuscript is now suitable for publication, and thank you in advance for allowing us to contribute to your journal. Please feel free to contact me with any questions regarding the manuscript.

Yours sincerely,

Xianqun Fan

M.D., Ph.D. Professor of Ophthalmology
Department of Ophthalmology, Ninth People's Hospital
Shanghai Jiao Tong University, School of Medicine
No. 639 ZhiZaoJu Road, Shanghai (200011), China
Email: fanxq@sjtu.edu.cn